

TECHNICAL NOTE

Forest Footprint for Cities

Methods for Estimating Deforestation and Associated CO₂ Emissions Embodied in Products Consumed in Cities

Mack Phillips, B. Scott Francisco, Sarah J. Wilson, Paige Langer, Larissa Jarvis, Noah Garcia

CONTENTS

Technical notes document the research or analytical methodology underpinning a publication, interactive application, or tool.

Version 1.0, December 2022

Suggested Citation: Phillips, M., B. S. Francisco, S. J. Wilson, P. Langer, L. Jarvis, and N. Garcia. 2023. "Forest Footprint for Cities. Technical Note. Washington, DC: World Resources Institute. Available online at: doi.org/10.46830/ writn.20.00128.

1. INTRODUCTION

With their high concentrations of people, influence, and consumption, cities have an enormous—and growing—impact on climate change. Through initiatives like C40 Cities, the Global Covenant of Mayors for Climate and Energy, the World Mayors Council on Climate Change, and the Race to Zero campaign, many cities are setting their own agendas to combat climate change. However, an acknowledgment that cities have substantial impacts on climate and the environment far outside their own borders is just starting to gain traction. Cities' actions, both inside and beyond their boundaries, will be key to meeting the climate action targets set by national and international agreements. New tools such as the Greenhouse Gas Protocol's Supplemental Guidance for Forests and Trees are essential to help communities understand how implementing nature-based solutions within their boundaries can contribute to achieving their climate goals.¹ The Forest Footprint for Cities can provide a complementary method for understanding how city actions impact trees and forests outside their boundaries.

Tropical forest loss is a major contributor to climate change, representing 8–11 percent of global greenhouse gas (GHG) emissions annually (Seymour and Busch 2016; IPCC 2022). Nature-based solutions, including halting deforestation and allowing degraded forests to recover, could contribute as much as 30 percent of the CO₂ mitigation necessary to achieve 1.5°C stabilization (Roe et al. 2019). However, global trends are moving in the opposite direction, with the tropics losing 11.1 million hectares of tree cover in 2020 (GFW 2022). Commercial production of widely used commodities, such as soy, beef, palm oil, and wood fiber, remains the leading driver of tropical deforestation (Curtis et al. 2018; Goldman et al. 2020). Due to the remoteness of tropical deforestation, it is often difficult for those living in cities to comprehend the magnitude of the issue or how they might be contributing to it.

The Forest Footprint for Cities methodology connects global estimates of tropical and subtropical deforestation linked to agricultural production to commodity-specific international trade and city consumption. The methods ultimately present the city's Forest Footprint in terms of hectares (ha) of embodied deforestation (ED) consumed and the associated CO₂ emissions. Here, ED is defined as deforestation that is attributed to tropical and subtropical forest being cut and replaced by a certain commodity. The omission of ED emissions from most cities' GHG inventories highlights the need for climate action planning to expand beyond the boundaries of city jurisdictions.²

The main goals of this approach are to raise awareness of the scale of the overall impact of urban consumption on tropical and subtropical forests and to help cities focus their climate mitigation actions on high impact commodities. The Forest Footprint for Cities is intended primarily for use by city government officials; however, secondary users may include nongovernmental organizations and corporations. The Forest Footprint shows

- the hectares of ED, by commodity, consumed within the city (ha/capita);
- the approximate area of deforestation linked to city consumption of those commodities (ha/yr); and
- the resulting carbon dioxide emissions from embodied deforestation (tCO₂/yr).

To make the Forest Footprint accessible to its target audiences, the output will be presented via the Forest Footprint for Cities Dashboard web application. The Dashboard will allow those without a technical background to understand the impact that city consumption has on tropical and subtropical forests through everyday consumption and to visualize how reducing the volume or the impact of specific commodities might change the footprint. The Forest Footprint for Cities methodology and Dashboard aim to present a compelling, data-driven call to action for cities to take steps to reduce their impact on tropical and subtropical deforestation.

This technical note explains the methods used to calculate cities' Forest Footprints. We use the term Forest Footprint in this technical note to refer to both the methodology and its outputs (i.e., hectares of deforestation and tonnes of CO₂ emissions (tCO₂) associated with consumption of ED). Section 2 provides an overview of methods used for calculating the Forest Footprint and outlines in more detail the data and calculations involved in each component. Section 3 explains how city-level consumption of the commodities included in the Forest Footprint methodology is estimated; and Section 4 describes how questions of attribution and traceability are addressed. Section 5 describes the final calculation of the Forest Footprint; and Section 6 outlines how GHG emissions associated with ED are calculated. Finally, Section 7 discusses comparisons, assumptions, and limitations.

2. OVERVIEW OF METHODS

The Forest Footprint for Cities focuses on tropical and subtropical deforestation, where deforestation is defined as the permanent land-use change from forest land to crop, pasture, or plantation forest land (Pendrill et al. 2019a, 2019b, 2020). While this includes some temperate regions (i.e., Chile, Australia, New Zealand, and Japan), the majority of temperate and boreal forests are excluded from this analysis as there is little permanent deforestation in these forests (Curtis et al. 2018). Further, what data exist on deforestation in temperate and boreal forests is not able to be parsed out into commodity-specific drivers of deforestation as in Pendrill et al. (2019a, 2019b, 2020). Temperate and boreal forest loss and degradation are important issues; however, the vast majority of recent forest loss in these areas is attributed not to food products (as with tropical and subtropical deforestation) but to timber harvesting in managed forests (Curtis et al. 2018; Goldman et al. 2020; Hoang and Kanemoto 2021). As forests that are harvested in temperate and boreal zones are likely to be replanted or naturally regenerated, this forest loss falls outside the scope of the Forest Footprint methodology.³

The Forest Footprint methodology is based on four main calculations. The first calculation is the annual city-level consumption of all commodities (given in tonnes per year) known to be linked to deforestation. The second calculation is the specific area of tropical and subtropical deforestation (given in hectares per year) associated with the production of each of those commodities (given in tonnes); the area is calculated through impact models. The third calculation is the city's total Forest Footprint, estimated by multiplying the city-specific consumption of each commodity (tonnes/year) by the associated impact models to produce a Forest Footprint (ha/year). Carbon dioxide emissions are then calculated based on the Forest Footprint calculations.



Figure 1 | Simplified Methodology of the Forest Footprint

Note: The arrows indicate how each of the three Footprints was calculated (e.g., the Global Average Footprint uses the global consumption calculation and the Global Production Model). The bottom part of the diagram shows how units were converted. *Source:* Authors.

2.1 Calculating annual rate of city-level consumption of commodities

The Forest Footprint methodology uses national-, urban-, and city-level consumption data, which allows for footprints to be calculated when data at higher levels of geographical specificity (e.g., city level) are unavailable. Commodities for which consumption data are calculated include beef, soy, palm oil, wood fiber, cocoa, coffee, rubber, sugar, and other crops. City-level consumption of these commodities is calculated in one of two ways based on data availability: absolute consumption (i.e., tonnes per capita of a specific commodity) or, when these data are unavailable, consumption ratios, which estimate the differences between subnational consumption and national consumption (e.g., dividing city-specific fuel expenditure by national fuel expenditure would provide a ratio with which subnational consumption of biofuel based on palm or soy oil can be calculated). Thus, the Forest Footprint provides cities with footprint estimates that can then be further refined.

2.2 Calculating area of tropical and subtropical deforestation associated with each unit of commodity consumed, or "impact models"

Data from the Land-Balance Deforestation Attribution Model (LanBaDA) are used to calculate the ED in both hectares and tonnes of CO₂ emissions associated with a unit of commodity produced (Pendrill et al. 2019a, 2019b, 2020). LanBaDA uses global land-use data from FAO (2020)-which is based on annual national surveys from FAO's Forest Resources Assessment and FAO's Land Use, Irrigation and Agricultural Practices questionnaire-and tropical and subtropical forest loss data from Hansen et al. (2013) to estimate the area of tropical and subtropical deforestation associated with the production of commodities. Based on these data, the Forest Footprint is able to utilize two different methods to attribute deforestation to consumption: the Global Production Model, which represents an even distribution of deforestation (both tropical and subtropical) across analyzed commodities produced globally; and the Physical Trade Model, which separates the consumption of commodities whose production is linked or not linked to tropical and subtropical deforestation through the use of international trade data.

Table 1 | Comparison of the Different Footprint Variants

FOOTPRINT NAME	FOOTPRINT DESCRIPTION	FOOTPRINT RELEVANCE AND LIMITATIONS		
Global Average	Consumption: This calculation assumes the equal consumption of commodities per person worldwide.	This calculation is a useful benchmark, as the per capita rate is the same worldwide. It provides an estimate that would be valid in a closed- system view where all consumption and deforestation is accounted for with no leakage.		
	Impact: This calculation uses the global average deforestation impact for each commodity (by weight or volume units), regardless of its country of production.	It does not account for the significant differences in consumption rates from one city to another or differences in production based on differing geographies or jurisdictions.		
Distributed Impact	Consumption: This calculation estimates the city's unique consumption patterns based on available national or subnational data. Impact: This calculation uses the global average deforestation impact for each commodity (by weight or volume units), regardless of its country of production.	By assuming the maximum liquidity of commodities traded in global markets (i.e., where increased consumption of American soybeans or beef in the U.S. increases global demand and incentivizes deforestation for soy farms and ranches in Brazil and Indonesia [see Box 1]), this calculation is better able to account for the impacts of city-specific consumption patterns on tropical and subtropical deforestation based on the assumption that global demand is not impacted by any individual city's consumption.		
Trade Flow	Consumption: This variant calculates consumption based on national-level commodity imports via the Physical Trade Model and subnational consumption data, then factors for city population. Note that in this unique case the impact model (Figure 2) affects how the consumption model calculates consumption. Impact: This calculation takes into account the country of origin of all commodities and traces the deforestation impact attributed to each using the Physical Trade Model.	This calculation begins with country-specific data on production, imports, and attributable deforestation of all commodities, thereby offering a high degree of specificity and accuracy on some commodities. It does not account for consumption that eludes commodity tracking, commodities that are embedded in other imports, or indirect deforestation due to global market liquidity.		

Source: Authors.

2.3 Calculating a city's Forest Footprint

The Forest Footprint estimate represents the ED in products consumed in a given city. It is based on an average of three distinct footprint variants—Global Average, Distributed Impact, and Trade Flow—each of which is calculated based on different combinations of consumption calculations and impact models (Figure 1). This flexibility allows cities to compare and contrast the impacts of their consumption on tropical and subtropical forests, based on the different assumptions of these models.

These footprints align roughly with the three footprints proposed by the European Commission (2013, 97) but are based on simpler methods of estimation. Together, these footprint variants are intended to give city staff and residents a more comprehensive understanding of their impact on tropical and subtropical forests. The different footprint calculations are summarized in Table 1.

As Figure 1 and Table 1 show, the Global Average Footprint calculation uses global consumption data, while the Distributed Impact and Trade Flow Footprints can use national-, urban-, or city-level consumption data. For the purposes of this technical note, "urban" refers to data representing all urban areas within a given country, whereas "city" refers to consumption data specific to a given city.⁴ City population is determined from Natural Earth's Populated Places data set, which provides city populations for the 500 most populous global cities.⁵

2.4 Calculating a city's CO₂ emissions estimate based on its Forest Footprint calculation(s)

CO₂ emissions are calculated as a range of per-hectare values from recent (2005–17) tropical and subtropical deforestation taken from Pendrill et al. (2020), who use aboveground biomass estimates from Zarin et al. (2016) and belowground biomass estimates from Mokany et al. (2006). The Forest Footprint methodology also includes emissions estimates that incorporate carbon opportunity costs (i.e., forgone carbon sequestration [see Searchinger et al. 2018; Waite et al. 2019]). Emissions from deforestation and carbon opportunity costs are presented separately to communicate the total annual CO₂ emissions from consumption of ED and wider agricultural land-use emissions for the city in question.

The following sections (3–6) describe each of these four calculations in more detail.

3. ESTIMATING CITY RESIDENTS' CONSUMPTION

The first step of the Forest Footprint calculation involves estimating the per capita consumption of key forest-risk commodities by city residents. At the finest level of geographic specificity (city level), the Forest Footprint calculates an average city resident's consumption profile for each commodity. This is the preferred form of consumption data (see Table 2), but data are not always available at this level of detail and resolution. Census and survey programs often report data by category, not by specific commodity (e.g., money spent on meat and meat products rather than beef), and many commodities are embedded within other products, making their actual consumption difficult to deduce (e.g., palm oil in processed foods and cleaning products).

Table 2 | Types of Consumption Data and Their Preferred Order of Use

DATA TYPE	METRIC	SCOPE	PREFERRED ORDER OF USE	EXAMPLE	
Direct commodity- specific consumption	Fresh weight (e.g., tonnes, kilograms per capita, etc.)	City and commodity specific data	1	Tonnes of coffee consumed in city A	
		Consumption data for all urban areas in country of interest	2	Tonnes of chocolate consumed in all urban areas of country B	
Consumption of product categories which include desired commodity but cannot be disaggregated ^a	Fresh weight (e.g., tonnes, kilograms per capita, etc.)	Commodity-specific city- and national-level data	3	1.0 kg of red meat ^b consumed per capita per year in country B and 1.2 kg of red meat consumed pe	
		Commodity-specific urban- and national level data	4	would produce a city-level ratio of 1.2	
	Data provided in metrics other than fresh weight (e.g., USD expenditure per capita)	Commodity-specific city- and national level data	5	US\$1000 spent on red meat per household per year in country B and \$1200 spent on red meat	
		Commodity-specific urban- and national level data	6	per household per year for city A (located in country B) would produce a city-level ratio of 1.2	
General consumption metrics which include desired commodity but cannot be disaggregated ¹	Data provided in metrics other than fresh weight (e.g., USD expenditure per capita)	Commodity-specific city- and national level data	7	\$2000 in total household expenditure per household per year in country B and \$2500 in	
		Commodity-specific urban- and national level data	8	total household expenditure per household in city A (in country B) would produce a city- level ratio of 1.25	

Notes:

a These data types are used to calculate consumption ratios, which make use of consumption data that are otherwise unable to be disaggregated into fresh weight of a commodity. Thus, both national- and subnational-level data are used to convert national-level direct consumption data to subnational levels.

b Following COICOP-based reporting, the red meat category includes both beef and many other forms of red meat (e.g., mutton, pork, etc.).

Source: Authors.

In instances where city-level consumption data are unavailable, a hierarchy of consumption ratios (calculated at different levels of geographic aggregation) is used to develop appropriate estimates based on the best data available. These consumption ratios are similar in nature to those used in the Ecological Footprint (Wackernagel et al. 2006; Rees and Wackernagel 2008), which were developed to analyze the consumption of energy, water, and food in cities from national-level data (see also Moore et al. 2013; Goldstein et al. 2016; Baabou et al. 2017). The preferred order of use for different types of consumption data within the Forest Footprint methodology is shown in Table 2.

3.1 Products Used to Calculate Consumption of Commodities

In order to calculate the quantity of commodities consumed within a city, it is necessary to understand the forms in which they are consumed. Here we identify some of the key products used to determine subnational consumption of each commodity either directly or via consumption ratios. This section also describes how the calculation of commodity consumption differs slightly when used in the Distributed Impact versus the Trade Flow Footprint variants (for the Global Average Model, no calculation is necessary since consumption under this model is the same for all countries and cities).

Box 1 | Consumption Calculations for Main Commodity Categories

Soy: Soy is both a food and a fuel source for humans. As a food source, it is consumed directly in products such as tofu and indirectly through consumption of meat and dairy that were raised on soy livestock feed consumption. Of the total global soy supply, approximately 13 percent is used for direct human consumption, 17 percent is used as a biofuel feedstock, and 70 percent is used as animal feed (Potts et al. 2014). Soybean oil is a common feedstock for biofuels, representing 27 percent of global biodiesel feedstock use (Steinweg et al. 2019).

While the Trade Flow Footprint can track soy meal consumption at the national level, we account for embodied soy meal consumed through meat and dairy in the Distributed Impact Footprint at the national level using a less direct method. As values of soy meal consumed per unit of livestock produced vary widely, national rates are used for the Distributed Impact Footprint when such data are available (e.g., IFIF 2022); in the event that appropriate figures are unavailable, the soy meal content of meat consumption is estimated based on average values from Table B-1, which give the amount of soy meal needed to produce one unit of livestock product for beef and veal, pork, poultry, eggs, and milk based on EU and U.S. data. When local feed conversion rates are not available, EU rates are used for European cities, the U.S. rates for American cities, and the EU-U.S. average rates for cities in all other countries. Estimates of soy meal consumption through meat and dairy products should be checked against national soy meal consumption data for accuracy; however, producers in certain countries may not use soy meal for feed at all. This method is used for calculating direct consumption at all levels for the Distributed Impact Footprint calculation and may also be used as a consumption ratio for the Trade Flow and Distributed Impact Footprint calculations.

Palm Oil: On a global scale, palm oil is used more than any other vegetable-based oil in food products, accounting for one third of global vegetable oil production (Potts et al. 2014). Palm oil is consumed directly in processed food, as an ingredient in household products like soaps and cosmetics, and in biofuels. As Pendrill et al. (2019a) and Proforest (2011) note, the highly processed nature of

these products makes estimating the consumption of palm oil difficult but not impossible. Palm oil is important as a feedstock for biofuels, representing 15.1 percent of global biofuel market volume share in 2019 (Grand View Research 2020) and 53 percent of the EU's palm oil imports (Oil World 2019).

Beef: Meat, leather, and dairy are three commodities associated with the deforestation caused by expansion of pastureland for cattle ranching; however, we include only meat in this category. Based on Brugnoli and Král' (2012) and Gac et al. (2014), Pendrill et al. (2020) attribute around 95 percent of ED to beef raised on pastureland. The remaining 5 percent is attributed to leather, as most leather comes

PRODUCT		SOY MEAL PER UNIT OF PRODUCT			
		EU	U.S.A.	EU-U.S. Average	
Beef and Veal		232 g/kg	77 g/kg	155 g/kg	
Pork		648 g/kg	508 g/kg	578 g/kg	
Poultry	Chicken	967 g/kg	788 g/kg	878 g/kg	
	Turkey	n/a	1079 g/kg	1079 g/kg	
Eggs		32 g/egg	28 g/egg	30 g/egg	
Dairy	Milk	21 g/L	37 g/L	29 g/L	
	Cheese	186 grams/kg	n/a	186 grams/kg	

TABLE B-1 | Soy Meal Consumption via Meat Products Used in the Distributed Impact Footprint (beef and veal, pork, and chicken)

Sources: E.U. data from van Gelder et al. (2008); U.S. data from Decision Innovation Solutions (2021) and FAO (2020).

from beef cattle. Following Pendrill et al. (2020), consumption of leather is not accounted for within the footprint, because trade in genuine leather products is difficult to trace, as it is often grouped with composite leather in data (e.g., by UN COMTRADE).

While dairy products are often also associated with other consumption-based carbon footprinting endeavors, following Opio et al. (2013), Pendrill et al. have not attributed any deforestation to dairy products, because they "assume that pasture expansion into forests is primarily for extensive cattle grazing for meat production (and not dairy products)" (2019b, 3). Dairy products in some cases may represent significant risk of increasing consumption of ED—for example, when dairy cattle are fed soy meal or palm kernel meal—however, the Footprint accounts for this through soy and palm oil consumption.

Wood Fiber: In 2015, industrial roundwood accounted for one third (429 million m³) of tropical and subtropical production of primary timber products, with fuelwood comprising the remaining two thirds (around 1 billion m³) (Held et al. 2021, based on FAO 2020). The Forest Footprint methodology considers wood used for construction, commercial fuelwood, and paper and paperboard. Plantation forestry has been a significant driver of deforestation in tropical and subtropical forests, where primary forest is cleared to establish fast-growing tree farms (Heilmayr 2014; Davis et al. 2015; van Straaten et al. 2015). Based on Pendrill et al.'s (2019a) LanBaDA model, only areas converted from unmanaged to plantation forests are identified as ED.

Cocoa, coffee, rubber, and sugar: These commodities have smaller impacts on tropical and subtropical deforestation compared to the previous commodities but are more commonly associated with tropical deforestation than the Other Crops category and have higher per unit impacts per unit of fresh weight. Globally, 92 percent of cocoa produced is exported, with European and North American markets consuming the majority (FAO n.d.). Coffee consumption is not dissimilar, with importing countries representing 70 percent of yearly global consumption (ICO 2020). Sugar is largely produced in tropical and subtropical countries, with around 22 percent in 2017-19 being used for ethanol production (OECD and FAO 2020). China is the world's largest consumer of natural rubber, where around 70 percent of natural rubber is used in tires and tubes (Statistics and Planning Department, Rubber Board 2018). Other products using natural rubber not accounted for by the Forest Footprint include latex, technological products, food storage, and adhesives. These four commodities are calculated and presented separately within the Forest Footprint.

Other Crops: This category represents crops that are not commonly associated with tropical and subtropical deforestation and therefore represent a small fraction of ED. These crops are mostly food products, as well as fiber crops used for clothing and household textiles. As the category represents more than 100 individual commodities, these have been grouped together in the Forest Footprint.

Consumption data are reported in various forms depending on the commodity and the data source and often require processing to transform raw data into the desired metric of tonnes consumed per capita per year. This is because commodities are consumed in different forms (e.g., palm oil can be consumed as food, fuel, and in household products), and consumption is measured in different ways (i.e., as money spent on a commodity or as nutritional intake in the form of a commodity). Further, the absolute quantity consumed (e.g., palm oil in soaps) is generally not accounted for directly, as the content of commodities in these products varies widely, is often difficult to calculate, and has little effect on city footprints. Instead, many of these product categories are captured in the calculation of subnational consumption ratios (see Section 3.4 for further explanation). Table 3 outlines the general breakdown of commodities, subcategories, and product categories used. Box 1 provides more detail on how consumption for each commodity was calculated.

3.2 Calculating National Absolute Consumption

In most cases, consumption of commodities is estimated at the national level to provide a reference for, or to represent, city-level consumption when city-level data are unavailable. The Distributed Impact Footprint calculates national apparent consumption using UN COMTRADE and FAOSTAT as primary sources of commodity-production, import, and export data. Apparent consumption is the sum of annual production and imports minus all exports; it differs from absolute consumption data in that it does not measure consumption directly (e.g., how much beef residents of a given country eat per year) but indirectly (e.g., how much beef is available for residents of a given country to eat per year after trade is accounted for). Commodity-specific consumption data from national agencies and other credible sources are used to validate these calculations; however, such data may not always be available. In circumstances where UN data differ from national factors of production, import, export, and consumption data may be averaged after confirming the comparability of measurements.

Table 3 | Products and Product Categories Indicating Consumption of Commodities

COMMODITY	COMMODITY Subcategories	PRODUCTS USED TO CALCULATE ABSOLUTE CONSUMPTION	PRODUCT CATEGORIES USED TO CALCULATE SUBNATIONAL CONSUMPTION RATIOS ^F	
Soy Food		Soybeans, ^a soybean oil ^{b, a}		
L	Livestock feed	Soy meal, beef, chicken, pork, milk, eggs, cheese $^{\scriptscriptstyle \mathrm{b}}$	Beef, ^{c,e} chicken, ^c pork ^c	
	Biofuel	Soybean oil, ^{b, e} biofuels ^{b, e}	Liquid fuels ^{c, e}	
Palm Oil	Food	Palm oil,ª palm kernel oil ª, b	Vegetable oils; $^{\rm c}$ other appliances, articles, and products for personal care $^{\rm c}$	
	Livestock feed	Palm kernel meal ^b	Beef ^{c, e}	
	Biofuel	Biofuels ^{b, e}	Liquid fuels ^{c, e}	
Beef	Food	Beef, meat of other bovine animals, ${}^{\mbox{\tiny D}}$ meat preparations of bovine animals ${}^{\mbox{\tiny D}}$	Live animals, meat, and other parts of slaughtered land animals ^c	
Wood Fiber	Wood for construction	Roundwood, $^{\rm b}$ sawnwood, $^{\rm b}$ and wood-based panels $^{\rm b}$	Construction, expenditure on wood products by construction sector	
	Fuelwood	$Fuelwood,{}^{\mathtt{b}}charcoal,{}^{\mathtt{b}}woodpellets{}^{\mathtt{b}}$	Solid fuels, $^{\circ}$ type of energy use by household	
	Paper and paperboard	Newsprint, paper, and paperboard ^d		
Rubber		Rubber (natural), ^a tires ^c	Purchase of vehicles ^c	
Сосоа		Cocoa paste, cocoa butter, cocoa powder, chocolate, cocoa drinks, ^ cocoa, and cocoa-based food products ^ $$		
Coffee		Coffee (green) ^b	Coffee and coffee substitutes ^c	
Sugar		Sugar ^b	Sugar, confectionery, and desserts, $^{\rm c}$ liquid fuels $^{\rm e}$	
Other Crops	Торассо	Tobacco (unmanufactured) ^{a, b}	Tobacco ^c	
Spic	Spices	The full list of products included in this category is provided on the	Spices, culinary herbs, and seeds ^c	
	Other cereals	Forest Footprint dashboard webpage. ⁹	Cereals and cereal products ^c	
	Rice			
	Roots and tubers		Tubers, plantains, and cooking bananas ^c	
	Pulses		Pulses°	
	Fiber crops		Clothing, ^c household textiles ^c	
	Oilseeds		Vegetable oils, ^c liquid fuels ^e	
Tree nuts			Fruits and nuts ^c	
	Fruits			
	Vegetables		Leafy or stem vegetables,° fruit-bearing vegetables,° green leguminous vegetables,° and other vegetables°	
	Crops nec ^h		Stimulants ^c and spices ³	

Notes:

a Category included under "Food" in the FAO New Food Balances database (FAO 2020). b Category from UN COMTRADE database (UN 2015).

c A COICOP category often used in household consumption surveys. Because these data are usually given in terms of monetary value, they are often better taken as consumption indicators than as products.

d To avoid double counting, we omit "Wood pulp" and "Chips and particles" otherwise included in Kastner et al.(2011a), as the Footprint considers final consumption of paper and not pulp.

e To avoid double counting, we use country-specific profiles to determine the attribution of product consumption (e.g., biofuels based on palm oil or soybean oil) to the appropriate commodities.

f These products and categories are used to calculate commodity-specific consumption ratios at subnational levels. See Section 3.4 for further detail.

g The Forest Footprint dashboard can be found at www.forestfootprint.org.

h "Not elsewhere classified."

Source: Authors.

National-level apparent consumption $(C_{{\it napp}})$ is calculated as follows:

(1)

$$C_{napp} = P_{nat} + I_{nat} - E_{nat}$$

Where P_{nat} , I_{nat} , and E_{nat} represent national production, imports, and exports of a given commodity, respectively.

The Trade Flow Footprint calculation, on the other hand, tracks the international trade of commodities from the country of production to the country of final consumption. National consumption is taken directly from Pendrill et al. (2020), which provides hectares of ED and the associated GHG emissions instead of fresh weight of commodities consumed. For both Distributed Impact and Trade Flow, national consumption provides a basis for subnational consumption ratios to be applied.

3.3 Calculating Subnational Absolute Consumption

The most accurate data for the purpose of the Forest Footprint methodology is per capita consumption of individual commodities for the city of interest. Types of data sources that report subnational data are household surveys (e.g., household income and spending, health, and nutrition), economic surveys, and research publications (e.g., food recall surveys). Data from these surveys and publications can be reported for specific cities, urban areas, or aggregated into total national urban areas. When data are reported for specific cities, per capita consumption for that city is calculated. When data are reported for urban areas, per capita consumption is calculated for the country's total urban population.

3.4 Calculating Commodity-Specific Consumption Ratios

For instances where city-specific or urban estimates of consumption are difficult to source in fresh weight or are included in product categories that are difficult to disaggregate by commodity (e.g., sugar, confectionery, and desserts), urban- and city-level estimates are calculated from national consumption using consumption ratios based on other commodity-specific data, such as household expenditure data or economic censuses. For example, the ratio of average annual household beef expenditure within a specific city to the national average expenditure is used to define the beef consumption ratio for that city. These ratios are used to scale national-level consumption data to the city or urban level (e.g., \$400 spent on vegetable oils per capita in country X and \$600 spent on vegetable oils per capita for city Y located in country X would allow direct or apparent national palm oil consumption to be brought to the city level by a factor of 1.5). Whenever possible, data sets that include both national and either urban or city figures are prioritized over combinations of different data sets (e.g., city-level data from one source and national from another), as this assures that data used in subnational conversions are as consistent as possible. The Commodity-Specific Consumption Ratio (CR_{com}) is calculated as follows:

(2)

$$CR_{com} = \begin{cases} \frac{Exp_{city}}{Pop_{city}} \div \frac{Exp_{nat}}{Pop_{nat}} \text{ if } Exp_{city} \text{ is available} \\ \frac{Exp_{urb}}{Pop_{urb}} \div \frac{Exp_{nat}}{Pop_{nat}} \text{ otherwise} \end{cases}$$

Where city-level and urban-level average household expenditure is given by (Exp_{city}) and (Exp_{urb}) , respectively, and is measured against average national expenditure (Exp_{nat}) and the relative size of the city population to the national population $(Pop_{city} \text{ and } Pop_{nal})$.

3.5 Calculating the Urban Expenditure Ratio

In cases where few data on commodity-specific city or urban consumption exist, the Forest Footprint uses a general urban consumption ratio (representing total consumption as opposed to commodity-specific consumption) to estimate urban-level consumption of a given commodity. The Urban Expenditure Ratio (EF_{urb}) is calculated as follows:

(3)

$$EF_{urb} = \frac{Exp_{urb}}{Pop_{urb}} \div \frac{Exp_{nat}}{Pop_{nat}}$$

Where EF_{urb} represents a ratio of urban and national total household consumption expenditure (Exp_{urb} and Exp_{nat}) at purchasing power parity provided by World Bank (2021b) or other international sources (OECD 2021 or EUROSTAT 2021). This ratio is taken from total household expenditure and considers differences in urban and national purchasing power parities. Thus, this ratio does not reflect the consumption of any one or group of commodities, but rather of all commodities and other products (e.g., electronics, medication, and clothes).

4. FOREST IMPACT MODELS

In estimating a city's Forest Footprint, it is necessary to determine the amount of ED (given in hectares) attributed to each unit (given in tonnes) of each crop consumed and address questions regarding the international trade of commodities that are inherent in the modeling. As the goal of the Forest Footprint is to show the impact of city consumption on tropical and subtropical forests, the Footprint considers only *tropical and subtropical* deforestation linked to the expansion of agricultural lands and forest plantations.

4.1 Attribution

To estimate the amount of tropical and subtropical deforestation linked to agricultural production and forest plantations globally, the Forest Footprint uses deforestation attribution data from Pendrill et al. (2020). These data build upon two previous studies (Pendrill et al. 2019a, 2019b), which provide more detailed methodological explanations. Pendrill et al.'s LanBaDA model is based on two main premises: first, "where cropland expands, it first expands into (converts) pastures (if there was a gross loss of pasture area) and then into forests (if there was a gross forest loss);" second, "where pastures and forest plantation areas expand they primarily replace forest land" (2019a, 3). Given these assumptions, the quantity of forest loss from cropland, pastures, and forest plantations is attributed to commodities (soy, palm oil, hotspot crops, and other crops; beef; and wood, respectively) "in relative proportion to their expansion in area" (Pendrill et al. 2019a, 4). This attributed deforestation is then amortized over five years of the commodity's production following the deforestation event (Pendrill et al. 2019b).6

Approximately 49 percent (on average) of tropical and subtropical forest loss (defined by Hansen et al. 2013) is not attributable to the above categories and is not necessarily considered permanent land-use change or deforestation (Pendrill et al. 2020). Within the Forest Footprint, the Other Forest Loss category represents the impact that humans have on tropical and subtropical forests outside of the ED attributed to the production of commodities-an important action point for cities globally. This category (defined in Pendrill et al. 2019a) includes the influence of a number of other drivers of deforestation, such as urban expansion and wildfires, as well as mining, infrastructure, oil and gas operations, and underreported land-use change unaccounted for in the previous categories. Because of the complexity of attributing Other Forest Loss to city-level consumption, we divide global Other Forest Loss by global population, and attribute it as a global per capita average to urban residents of any city in our methodology.

4.2 Traceability and Impact Models

The Forest Footprint employs two different approaches to trace the deforestation from the source of production to the source of consumption:

- The first approach is based on the Global Production Impact Model, which attributes tropical and subtropical deforestation to global production of a given commodity, regardless of where the commodity was sourced. This allows for the capture of indirect effects that the global market produces (e.g., the pressure of U.S. consumption of soy on Brazilian or Indonesian soy expansion) that are difficult to capture in detailed trade models. This approach is used for the Global Average and Distributed Impact Footprints.
- The second approach builds on Pendrill et al.'s (2019a) Physical Trade Model, which is used only in the Trade Flow Footprint. This model tracks national-level consumption of commodities from specific source regions with their respective differing deforestation factors.

At the national level, the Global Production Impact Model uses a simplified set of assumptions in calculating the hectares of ED associated with the consumption of a given commodity. Tropical and subtropical deforestation per commodity is summed and divided by total global production to create a Forest Impact Factor (e.g., the Forest Impact Factor of soy is 1.46 hectares per tonne, meaning an average of 1.46 hectares are deforested in tropical and subtropical regions for every tonne of soy produced in dry weight based on a five-year amortization rate). This is given in the equation below:

(4)

$$FI_{com} = ED_{com} \div P_{com}$$

Where FI_{com} represents the Forest Impact Factor, ED_{com} represents the total hectares of ED linked to a given commodity using the Global Production Model, and P_{com} represents the total annual global production of the given commodity and its secondary products. The Global Production Model assumes that all soy—regardless of the location of its consumption or production—is linked to the same amount of deforestation (see Box 2).

Table 4 provides an overview of the total deforestation linked to each commodity and its resulting Forest Impact Factor as calculated through the Global Average Model. We chose to use ED based on a yearly average from 2005–17 as opposed to 2017 alone as the average represents a better basis for informing decisions due to yearly fluctuations in ED.

Table 4 | Deforestation Impacts of FootprintedCommodities, Based on the Global Production Model

COMMODITY		TOTAL TROPICAL AND SUBTROPICAL ED, 2005-17 AVG. (kha°)	TOTAL GLOBAL PRODUCTION, 2005-17 AVG. (Mt ^b)	FOREST IMPACT FACTOR, 2005-17 AVG. (m ² /t)°
Soybear	าร	390	267	14.6
Palm Oil		423	51	82.9
Beef		1,984	68	291.8
Wood Fi	ber	598	2,534	2.4
Сосоа		59	4.5	132.2
Coffee		40	8.5	47.0
Rubber		54	11.8	45.4
Sugar		37	1,981	0.2
	Rice	221	709	3.1
	Other cereals	444	1898	2.3
	Oilseeds nec	214	238	9.0
	Pulses	146	72	20.0
	Roots & tubers	226	779	2.3
Other	Vegetables	52	946	0.6
Crops	Fruits	57	761	0.6
	Tree nuts	5	14	3.5
	Fiber crops	2	77	0.3
	Crops nec ^d	31	24	12.6
	Other Crops Total	1,426	5,519	2.6
Other Forest Loss		3,911	-	5.1 ^e

^a kha = kilohectares.

^b Mt = megatonnes.

^c For the purposes of this table, the Forest Impact Factor, while normally calculated in hectares per tonne, has been presented in square meters per tonne to improve legibility and comprehension.

^d Pendrill et al.'s (2019b) category of the same name includes cocoa, coffee, and rubber, which have been separated in the Forest Footprint. Remaining in this category are tobacco and spices.

^e The Forest Impact Factor for Other Forest Loss is given here in square meters per capita. Sources: Tropical and subtropical data were derived from Pendrill et al. (2020) and global production data from FAOSTAT.

The Physical Trade Model uses FAOSTAT international trade data on roughly 400 items to trace the deforestation associated with the production of commodities to the country of their consumption (Pendrill et al. 2019b). The model is based on studies by Kastner et al. (2011b, 2014a), which convert products into "physical units of primary commodity equivalents" (e.g., 400 g cocoa per 1 kg of chocolate) to avoid double counting. This ensures that commodities that are processed in country A and exported as different products to country B (e.g., cocoa beans processed into chocolate) are discounted by the correct ratio from country A's total consumption of ED (since the chocolate would be consumed in country B). To determine a country's final consumption of ED, Pendrill et al. (2020) then convert the "physical units of primary commodity equivalents" associated with tropical and subtropical deforestation into the area of ED via FAOSTAT crop yield data (given in tonnes per hectare).

The Physical Trade Model has certain advantages over the Global Average Model; for example, it better accounts for trade in commodities embodied in certain products, such as bread and pasta, but not heavily processed products, such as frozen pizzas and soaps.⁷ At the same time, as Pendrill et al. (2019b) acknowledge, the Physical Trade Model estimates that countries that import livestock feed products (e.g., soy meal) consume more ED than countries that import meat products from livestock that grazed on these feed products; however, the authors claim the discrepancy is marginal. Furthermore, there remain significant differences in the results of different trade model methodologies (e.g., FAOSTAT, Multi-Regional Input Output (MRIO) modeling, and physical trade modelling), which are not yet fully understood (Kastner et al. 2014b). It is important to note when analyzing a city's Forest Footprint that this model should not be considered superior-but rather complementary-to the Global Production Model.

5. CALCULATING FOREST FOOTPRINT VARIANTS

The Forest Footprint is calculated using three variants. Each variant taken on its own gives city officials the opportunity to understand the different assumptions used to calculate their Forest Footprint. When averaged, the footprint variants present the city's impact in a simpler and more harmonized fashion. This supports the main goal of the footprint—namely, to raise awareness and empower city leaders and officials to support tropical and subtropical forests. In doing so, cities can avoid some of the common pitfalls of acting on a footprint (Chapman et al. 2017). This aligns with Pendrill et al. (2020), who warn that "the estimates of deforestation embodied in import should therefore be interpreted as a measure of deforestation risk."

The Forest Footprint variants may be assessed using consumption data at different levels of geographical specificity, as available: global, national, urban, and city. While each level gives a city-specific footprint, per capita consumption rate estimates may differ depending on whether they are calculated with global-, national-, urban-, or city-level averages. The first two footprints—Global Average (equation 5) and Distributed Impact (calculated at three different scopes in equations 6–8)—give footprints to cities based on the Global Average Model, while the Trade Flow Footprint (also calculated at three different scopes in equations 9-11) is based on the Physical Trade Model.

5.1 The Global Average Footprint

The Global Average Footprint assumes equal consumption of commodities per person worldwide and equal distribution of deforestation impacts. This footprint variant is used primarily as a benchmark since cities worldwide are assigned equal ED values per capita. The Global Average Footprint variant is calculated as follows:

(5)

$$FF_{ga} = (ED_{gtot} + FL) \times \frac{Pop_{city}}{Pop_{global}} \times \frac{Exp_{nat}}{Exp_{global}}$$

Where FF_{ga} represents the Global Average Footprint calculated at a national level irrespective of how much of each commodity is consumed in each city or country; *FL* represents Other Forest Loss; and ED_{gtol} represents the sum of ED attributed to all commodities. Exp_{nal} represents average national household expenditure given in constant 2010 US dollars taken from the World Bank (2021b), and Exp_{global} represents the average global national household expenditure from the same source (see Table 5 for the units associated with each variable).

5.2 The Distributed Impact Footprint

The Distributed Impact Footprint considers differences in consumption among cities while still assuming equal distribution of deforestation impacts regardless of country of production. The following equations are used to calculate the Distributed Impact Footprint variant at the national, urban, and city level, in that order:

(6)

$$FF_{din} = (\sum_{com} (FI_{gcom} \times C_{napp}) + FL) \times \frac{Pop_{city}}{Pop_{nat}}$$

Where FF_{din} represents the Distributed Impact Footprint calculated at the national level, and Σ_{com} represents the sum of ED consumed across all commodities. This is given by multiplying Forest Impact Factor (FI_{gcom}) of a given commodity by the national apparent consumption (C_{napp}) of the same commodity. Other Forest Loss (FL) is added to the total (Σ_{com}) and scaled from national (Pop_{nat}) to city population (Pop_{city}).

(7)

$$FF_{diu} = ((\sum_{com} (FI_{gcom} \times C_{napp}) \times EF_{urb}) + FL) \times \frac{Pop_{city}}{Pop_{nat}}$$

Where FF_{diu} represents the Distributed Impact Footprint calculated at the urban level, and Σ_{com} represents the sum of ED consumed across all commodities. This is given by multiplying the Forest Impact Factor of a given commodity (FI_{gcom}) by the national apparent consumption (C_{napp}) of the same commodity and the Urban Expenditure Factor (EF_{urb}) . Other Forest Loss (FL) is added to the total (Σ_{com}) and scaled from national (Pop_{nat}) to city population (Pop_{city}) .

(8)

$$FF_{dic} = \begin{cases} (\sum_{com} (FI_{gcom} \times C_{com}) + FL) \times \frac{Pop_{city}}{Pop_{nat}} & \text{if } C_{com} \text{ is available} \\ (\sum_{com} (FI_{gcom} \times C_{napp} \times CR_{com}) + FL_{other}) \times \frac{Pop_{city}}{Pop_{nat}} & \text{otherwise} \end{cases}$$

Where FFdic represents the Distributed Impact Footprint calculated at the city level, and ∑com represents the sum of ED consumed across all commodities. This is given by multiplying the Forest Impact Factor of a given commodity (FIgcom) by either the subnational direct consumption (Ccom) of the same commodity—if available—or the national apparent consumption (Cnapp) and the Commodity-Specific Consumption Ratio (CRcom) of the same commodity. In both cases, Other Forest Loss (FL) is added to the sum before scaling from the national (Popnat) to city level (Popcity).

5.3 The Trade Flow Footprint

Finally, the Trade Flow Footprint traces deforestation attributed to commodities from their country of production to the country where they are consumed. Whereas the second Distributed Impact variant allows for the indirect effects of consumption on deforestation (e.g., increased demand for soybeans in the U.S. incentivizing deforestation for soy farms in Brazil), the third Trade Flow variant includes only the direct effects of consumption on deforestation (e.g., thus consumption of Brazilian-grown soy carries ED and Canadian-grown soy does not). The Trade Flow Footprint variant calculations at the national and city levels are given by the following equations: (9)

$$FF_{tfn} = \left(\sum_{com} ED_{pcom} + FL\right) \times \frac{Pop_{city}}{Pop_{nat}}$$

Where FFtfn represents the Trade Flow Footprint calculated at a national level, and Σ com represents the sum of ED consumed across all commodities. This is given by the ED consumed at a national level for each commodity (EDpcom)—as calculated by the Physical Trade Model—and Other Forest Loss (FL) before scaling from the national (Popnat) to city level (Popcity).

(10)

$$FF_{tfu} = \left(\sum_{com} ED_{pcom} \times EF_{urb} + FL\right) \times \frac{Pop_{city}}{Pop_{nat}}$$

Where FF_{ifu} represents the Trade Flow Footprint calculated at an urban level indifferent to commodity-specific consumption, and \sum_{com} represents the sum of ED consumed across all

Table 5 | Units Associated with Variables Used inFootprint Variant Calculations

VARIABLE	SHORTFORM	UNIT	
Global Average Footprint	FF _{ga}		
Distributed Impact Footprint (national, urban, city)	$FF_{din}FF_{diu}FF_{dic}$		
Trade Flow Footprint (national, urban, city)	$FF_{tfn}FF_{tfu}FF_{tfc}$	Hectares (ha)	
Other Forest Loss FL			
Embodied deforestation consumed at a national level	ED _{pcom}		
Forest Impact Factor	Fl _{gcom}	Hectares per tonne (ha/t)	
Subnational direct consumption	C_{com}	Toppoo	
National apparent consumption	C_{napp}	TOTITIES	
Population (national, city)	Pop _{nat} Pop _{city}	Persons	
Urban Expenditure Ratio	EF _{urb}		
Commodity-Specific Consumption Ratio	CR_{com}	none	

Source: Authors

commodities. This is given by multiplying the embodied deforestation consumed at a national level (ED_{pcom}) —as calculated by the Physical Trade Model—by the Urban Expenditure Ratio (EF_{urb}) , and *FL* represents Other Forest Loss before scaling from the national (Pop_{nar}) to city level (Pop_{cir}) .

(11)

$$FF_{tfc} = (\sum_{com} (ED_{pcom} \times CR_{com}) + FL) \times \frac{Pop_{city}}{Pop_{nat}}$$

Where FFtfc represents the Trade Flow Footprint calculated at a city level, and Σ com represents the sum of ED consumed across all commodities. This is given by multiplying the embodied deforestation consumed at a national level (EDpcom)—as calculated by the Physical Trade Model—by the Commodity-Specific Consumption Ratio (CRcom). Other Forest Loss (FL) is added before scaling from the national (Popnat) to city level (Popcity).

6. CALCULATING CO₂ EMISSIONS FROM CONSUMPTION OF EMBODIED DEFORESTATION

The final step of the Forest Footprint is calculating CO2 emissions from consumption of land-based products. This is given in two different metrics: CO2 emissions from consumption of recent tropical and subtropical ED and carbon opportunity costs (COC). The former represents CO₂ emissions from loss of aboveground and belowground biomass due to deforestation, whereas the latter represents the total historical amount of land-based carbon lost from vegetation and soils on productive agricultural lands for a given commodity (this quantity also represents the amount of carbon that could be stored if land in production were allowed to return to native vegetation). Both metrics are presented within the Forest Footprint to understand the relationship between CO₂ emissions from the consumption of ED and the larger, historical emissions of land-use conversion for agriculture; however, currently only the former is directly compatible with cities' GHG accounting methods (WRI et al. 2022).

The Forest Footprint uses deforestation emissions data from Pendrill et al. (2019b, 2020), given in tCO₂/ha. Pendrill et al. (2019b, 2020) give CO₂ emissions linked to ED as a single sum without a range or standard error. We contend that the single figure is appropriate in strongly communicating the measure of deforestation risk in terms that cities can relate to (see Table 7 and Section 7.3 for further discussion). It should, however, be acknowledged that emissions calculated from land-use change models are uncertain (Bontinck et al. 2020). This is largely due to the fact that tropical and subtropical forest carbon stocks themselves are difficult to ascertain (Ramankutty et al. 2006; Pan et al. 2011). The deforestation emissions for the Global Average, Distributed Impact, and Trade Flow Footprint variants (respectively, DE_{ya} , DE_{di} and DE_{ti}) are calculated according to:

(12)
$$DE_{ga} = FF_{ga} \times \frac{DE_{tot}}{t}$$

(13) $DE_{di} = FF_{di} \times \frac{DE_{tot}}{t}$

(14)
$$DE_{tf} = \sum_{com} DE_{com} \times \frac{Pop_{city}}{Pop_{nat}}$$

Where DE_{com} represents the total emissions from deforestation given by Pendrill et al. (2020), and *t* represents the number of years where deforestation has been assessed by Pendrill et al. (2020). The deforestation emissions for the Trade Flow Footprint variant (DE_{ty}) are calculated differently; the Physical Trade Model (Pendrill et al. 2020) traces deforestation emissions from the source country of deforestation to the country of final consumption. Thus, Σ_{com} represents the sum of deforestation emissions for each commodity (DE_{com}) consumed by the country in which the city is located. These calculations are to be performed with data from the closest level of geographic scope available (i.e., city level).

Carbon opportunity costs (COC) represent the total historical amount of land-based carbon lost from vegetation and soils on agricultural lands (Searchinger et al. 2018; Waite et al. 2019). COC factors for commodities are obtained by dividing the time-discounted annual rate of total historical carbon losses on lands producing a given commodity (e.g., beef, soy, etc.) by the total annual production of that commodity. The concept of time discounting means that CO₂ is attributed to the production of crops at a discounted (i.e., decreasing) yearly rate starting from the initial land-use conversion event. This metric considers carbon losses in both tropical and temperate ecosystems. Because carbon losses occur quickly but lands can be productively used for many years, the COC is annualized using a discount rate of 4 percent that effectively distributes the carbon losses from land conversion over 30-35 years. COC per kilogram of a commodity can also be thought of as the amount of carbon likely to be lost from plants and soils as a result of producing an additional kilogram of that food. COC is presented in tandem with GHG emissions from recent deforestation as a means of

Table 6 Categorical Connection of Carbon Opportunity Cost Factors to Forest Footprint Commodities

COMMODITY	COMMODITY Subcategory	ASSOCIATED COC Category	CARBON OPPORTUNITY COST FACTOR (KG CO, E/KG FRESH WEIGHT)
	Food	Soybeans/Tofu	5.85
Soy	Feed	Soy meal	4.92
	Fuel	Soybeans (oil)	9.98
Dalm Oil	Food	Palm (oil)	8.26
Pallill Ull	Fuel	Palm (oil)	8.26
Beef	Food	Beef and buffalo meat	143.92
	Wood		
Wood Fiber	Wood fuel	n/a	
	Paper		
	Сосоа	Сосоа	40.36
	Rubber	n/a	
	Coffee	Coffee	31.13
Hotspot Crops	Торассо	Stimulants and spices misc.	44.63
	Spices	Stimulants and spices misc.	44.63
	Sugar	Sugars and sweeteners	1.75
	Tree nuts	Tree nuts and seeds	7.59
	Pulses	Beans and pulses (dried)	10.53
	Fiber crops	Seed cotton	2.86
	Oilseeds nec	Vegetable oils	4.79
Other Crops	Rice	Rice	2.61
	Other cereals	Grains/cereals (except rice)	2.02
	Roots and tubers	Roots and Tubers	1.12
	Fruits	Fruits	1.03
	Vegetables	Vegetables	0.71

Sources: Authors and Waite et al. 2019.

showing the more general CO₂ impacts of the consumption of food products, rather than just impacts on tropical forests. COC is calculated as:

$$COC_{din} = \sum_{com} (COC_{com} \times C_{napp}) \times \frac{Pop_{city}}{Pop_{nat}}$$

Where COC_{com} represents the commodity-specific carbon opportunity cost factor (given in tCO₂ per tonne of commodity—see Table 6) and COC_{din} represents the total carbon opportunity cost calculated for the Distributed Impact Footprint variant at the national level. This equation is representative of similar COC calculations, which can also be done with equations 5–8 for the Global Average and Trade Flow Footprint variants by replacing FI_{com} with COC_{com} .

Because the deforestation emission factors from Pendrill et al. (2020) consider emissions from recent tropical and subtropical deforestation attributed to all the aforementioned commodities, and the COC considers emissions linked to historical land-use change across only food-based agricultural commodities, COC estimates are higher. For instance, the COC factor per kilogram of coffee is almost 400 times larger than the emission factor from recent coffee-related deforestation, while the COC factor is about 10 times higher per kilogram of beef and soy and 1.5 times higher per kilogram of palm oil, versus the respective emission factors from recent deforestation. As Pendrill et al. (2020) do not currently provide data on weights of commodities consumed through the Physical Trade Model, only the Global Average and Distributed Impact Footprints are able to be converted to COC. It is also important to note that COC estimates of wood and rubber products have yet to be developed and the consumption of wood fiber and rubber is therefore excluded from the COC estimates.

7. COMPARISONS, ASSUMPTIONS, AND LIMITATIONS

The Forest Footprint methodology relies on assumptions that must necessarily be made due to a lack of specific data and/ or for ease of comparability between cities. These assumptions are categorized by attribution, consumption, and deforestation (Sections 7.1–7.3) and represent limitations (Section 7.4) in the accuracy of the Forest Footprint methodology. Some of these limitations are inherent in the data sets we have selected, while others arise from assumptions specific to our methods, which we intend to address in future iterations of this methodology.

7.1 Comparison of Attribution Methods and Assumptions

As discussed in Section 4, Pendrill et al. (2020) rely on FAO land-use data to calculate amounts of ED. Deforestation attribution data have also been estimated via remote sensing from satellite imagery based on Hansen et al. (2013). A study by Curtis et al. (2018) estimated global drivers of deforestation using a forest loss classification decision-tree model via remote sensing. This improved upon the Hansen et al. (2013) data set, which estimated only tree cover loss and not deforestation.⁸ While this method is positioned to overcome many of the limitations of Pendrill et al.'s (2020) LanBaDA method, it is not used in the Forest Footprint calculations because it was developed at a resolution (10-km cells) that is too coarse for this methodology.⁹

Goldman et al. (2020) have further built upon the Hansen data by comparing spatially explicit GIS agricultural land-use data sets against Hansen et al.'s (2013) forest-cover-loss data to estimate "agriculture-linked deforestation." The study analyzes many of the same commodities as Pendrill et al. (2019a, 2019b, 2020), but the results from Goldman's and Pendrill's studies differ in important ways. Beyond the obvious methodological differences, Goldman et al. attribute discrepancies to their more fine-grained categorization of land uses, which, in turn, led to differences in ED estimates. For instance, Goldman et al. (2020) consider only cuts from forest plantations labeled as "wood fiber" from Harris et al. (2019) to be deforestation, whereas Pendrill et al. (2019, 2020) consider cuts from all forest plantations to be deforestation. ED from wood fiber, therefore, differed by a factor of four between the two studies.

For the purposes of attributing ED in this model, we chose to use Pendrill et al. (2019a, 2020). Because the Forest Footprint's main goal is to raise awareness around ED consumed in cities, our model benefits less from the detailed subnational and spatially explicit ED data provided by Goldman et al. 2020 than from a smaller set of assumptions covering a wide range of commodities that can be more quickly understood by a broader, non-specialist audience.

7.2 City Consumption Assumptions

All footprint calculations assume an average resident profile for each city being footprinted. While there are large discrepancies in individual consumption behavior (depending both on a resident's socioeconomic situation and personal choices), we again opted to provide an easily communicable and relatable snapshot of the impacts of city dwellers on tropical and subtropical deforestation.

As for the product categories used to calculate consumption ratios, it is assumed that within a given country there is no difference between products consumed in urban, non-urban, and city areas (e.g., average cocoa content in chocolate). Additionally, in certain cases, these commodity-specific consumption ratios may include items that do not contain any of the identified commodity (e.g., using vegetable oils taken from a household consumption survey to calculate the city-level consumption ratio of palm oil). As it is difficult to ascertain the consumption shares of commodities within all product categories (e.g., how much palm oil is used per kg of chocolate or soap), it is assumed that any difference in consumption of product categories (e.g., from national- to city-level consumption of vegetable oils) is directly proportional among all commodities within the product category. For example, while per capita vegetable oil consumption in country X might be higher than city Y (within country X), without knowing the contents of vegetable oil or the types of oil included in the survey it is impossible to estimate how much less palm, sunflower, or olive oil residents of city X consume compared to citizens of country Y.

Many of the data sets used for commodity-specific consumption data are based on different years (e.g., 2015 timber consumption and 2011 fuelwood consumption), whereas the forest impact factors are based on average deforestation linked to agricultural production over the 2005–17 period. No adjustments have been made to correct for the differences in consumption data by year; thus, it is assumed that consumption levels have neither increased nor decreased substantially since the data were published.

7.3 Deforestation Attribution Assumptions

Other Forest Loss assumes that forest loss (including deforestation) due to wildfires, forestry outside of plantation forests, shifting agriculture, and urbanization can be attributed to human actions and are thus equally the responsibility of people worldwide, who should take mitigating action regardless of their location or consumption patterns.

Estimates of GHG emissions from recent deforestation and forest loss vary widely. This is due to many factors, including geographical scope and other methodological differences. The

Table 7 | Comparison of Estimated Emissions from Tropical and Subtropical Deforestation

STUDY	EMISSIONS FROM DEFORESTATION (PG C/YR®)	TIME PERIOD	DEFORESTATION (MHA/YR ^D)	GEOGRAPHICAL Scope	GHG EMISSIONS PER HA OF Deforestation (TCO ₂ /HA)
Le Quéré et al. 2009. "Trends in the Sources and Sinks of Carbon Dioxide."	1.5 ± 0.7	1990—2005	8.351°	Global	659 ± 307
Pan et al. 2011. "A Large and Persistent Carbon Sink in the World's Forests."	1.30 ± 0.47 (net)	1990—2007	8.546 ^d	Tropical and subtropical	558 ± 301
Baccini et al. 2012. "Estimated Carbon Dioxide Emissions from Tropical Deforestation Improved by Carbon-Density Maps." ^e	1.14 (gross)	2000—10	7.933 ± 0.817 ^f	Dry and humid tropical	527 ± 48.9
Pendrill et al. 2020. "Deforestation Risk Embodied in Production and Consumption of Agricultural and Forestry Commodities 2005-2017."e	0.71 (net)	2005—17	4.974	Tropical and subtropical	519
Achard et al. 2014. "Determination of Tropical Deforestation Rates and Related Carbon Losses from 1990 to 2010." ^e	0.88 ± 0.36 (gross) ^g	2000—10	7.62 ± 0.33	Tropical	423 ± 292

Notes:

a Pg C/yr = Petagrams carbon per year.^b Mha/yr = million hectares per year.^c Le Quéré et al. (2009) take net global changes in forest from FAOSTAT's Global Forest Resources Assessment (FAO 2005).^d Pan et al. (2011) take net tropical and subtropical changes in forest area from FAOSTAT's Forest Resources Assessment (FAO and JRC 2010).^e Study uses carbon-density maps. ^fFor years 2000–05, Baccini et al. (2012) take tropical forest loss data from Hansen, et al. (2010).^e This range is based on minimum and maximum carbon density from Saatchi et al. (2011). Source: Authors. Forest Footprint and Pendrill et al. (2019b, 2020) focus on gross emissions from deforestation, whereas many other studies expand this scope to include forest degradation and reforestation (Seymour and Busch 2016, 41). Of the other studies that focused on deforestation and not degradation, Pan et al. (2011) was the only one able to determine net deforestation, whereas Baccini et al. (2012) and Achard et al. (2014) measured remotesensed forest loss. While Le Quéré et al. (2009) used land-use data to estimate emissions, the study does not disaggregate emissions from land-use change (for 2005–10, the study uses forest fire emissions as a proxy for emissions from deforestation) and does not distinguish deforestation by forest biome. As evident in Table 7, Pendrill et al.'s (2020) estimate of GHG emissions per hectare of deforestation lies within the ranges given by these studies.

7.4 Other Limitations

Many products are not fully captured within the footprint. In cases where countries import highly processed products of commodities (e.g., frozen pizzas), all footprint calculations may be underestimates due to the difficulties of calculating consumption of commodities embedded in categories of imported products. In other cases, commodities cannot be fully identified; for example, UN COMTRADE groups both real and imitation leather products together in its trade data. A different kind of problem exists in that no footprint calculation accounts for the impact of activities with little direct economic measure, such as subsistence farming and wood harvesting, which are nevertheless relevant to the consumption of products.

In addition, the Forest Footprint does not fully capture carbon opportunity costs related to animal products beyond beef (i.e., lamb, pork, poultry, eggs, and dairy products) as it captures only the carbon opportunity costs associated with soy-based animal feeds but not those costs associated with cropland or pastureland use beyond beef and soy production. It also does not capture carbon opportunity costs related to rubber or wood production.

There are many more effects of land-use change outside of the Forest Footprint that should be accounted for in promoting changes in consumption from one commodity to another (e.g., conversion of grasslands and degraded soil health). For instance, while palm oil is a major driver of tropical and subtropical deforestation, it has one of the highest yields (in tonnes per hectare) of all oilseeds. Thus, while switching vegetable oil or biofuel consumption to a different oilseed may result in a lower Forest Footprint it may result in other adverse effects such as increased or intensified agricultural land use in other environments, which may, in turn, lead to more deforestation than using palm oil. It is known that UN trade and production data (COMTRADE and FAOSTAT) present some inaccuracies in estimating apparent consumption at the national level-most notably in commodity misinvoicing in developing countries (Ndikumana 2016). Misreporting of trade volume is a well-known strategy used by transnational companies to avoid export tariffs on primary commodities; however, this is most common with high-value, low-weight products and goods without standard international pricing regimes. For example, the recent UNC-TAD report (Ndikumana 2016) identified cocoa from Côte d'Ivoire as a leading example of large-scale under-invoicing, along with Chilean copper, Nigerian oil, and South African gold. This means that commodity trade data may sometimes be underreported. On the production side, this phenomenon has been accounted for in certain cases by incorporating other comparably reliable sources of production data, such as U.S. Foreign Agriculture Service data; however, due to the number of commodities and their products assessed, the correction is often limited to outliers. Both discrepancies affect apparent consumption estimates and render the Distributed Impact Footprint estimate marginally higher than it might otherwise be.

The commodity-specific consumption data rely heavily on household survey data. While these data sets are widely available for most countries, they present some limitations and vary in their ability to capture current trends in commodity consumption (FAO and World Bank 2018). Survey methods may not adequately represent all consumption pathways, such as food that is acquired but not purchased and food that is consumed away from home (for instance, food consumed in restaurants is not accounted for). These considerations are especially important in developing countries where consumption patterns are rapidly changing. In addition, household survey data often report total household expenditure (in local currency, USD, or PPP), and geographically specific differences in commodity prices (such as rural and urban beef prices) are difficult to account for due to data availability and quality.

The use of household surveys also excludes the measure of cityspecific consumption of private businesses. While this may be a significant factor (and is accounted for nationally), it is difficult to attribute the consumption or waste of private businesses to cities as it would be necessary to know whether the businesses' products were being finally consumed within the city or not. For commodities such as wood fiber, economic census data can be used, which includes households, private corporations, and public institutions.

7.5 Summary

The Forest Footprint can help cities understand their impact on tropical and subtropical deforestation by presenting consumption data in an accessible manner. As the impacts of decisions made by city or metropolitan jurisdictions on tropical and subtropical forests are difficult to see, the Forest Footprint brings these impacts into the visible realm and sets the stage for further incorporation of consumption-based emissions and impacts in city governance and sustainability planning. By understanding their forest footprints, cities worldwide can join the growing number of private sector and national government actors who are reducing tropical and subtropical deforestation as they measure, monitor, and ultimately transform the infrastructure and the culture of urban consumption.

ENDNOTES

- 1. The guidance can be found here: https://ghgprotocol.org/gpc-supplemental-guidance-forests-and-trees.
- 2. At the time of writing, it is not required by the Global Protocol for Community-Scale Greenhouse Gas Emission Inventories (GPC) standards to include GHG emissions from consumption of products that embody deforestation linked to agricultural production—let alone any extraterritorial consumption-based emissions—within either the GPC BASIC or BASIC+ schemes (WRI, C40, and ICLEI 2021). We use the term "consumed embodied deforestation" as shorthand for "consumption of products whose production is associated with the loss of forest cover in tropical or subtropical zones" thus, deforestation is embodied in these products.
- 3. Curtis et al. (2018) attribute 0 and 1 percent of forest cover loss between 2001 and 2015 to permanent deforestation in Europe and North America, respectively. When portions of boreal or temperate forests are cleared for timber, they are almost always allowed to grow back, whether they are replanted or allowed to regenerate naturally. While this can have negative biodiversity and carbon impacts in boreal and temperate regions, these impacts are not as acute or as well understood as those of tropical and subtropical forests. City staff should be aware of these issues when choosing which building materials to promote (or procure) within the city. For all commodities apart from wood and rubber, temperate and boreal deforestation is indirectly represented in the carbon opportunity cost metric (see Section 6), which factors the global climate impact of dedicating land to production of the given commodity.

- 4. In certain cases where climate planning is carried out primarily at the metropolitan level (i.e., comprising multiple municipalities), a metropolitan footprint may prove to be more appropriate and must reflect not only the metropolitan population but also metropolitan consumption data.
- The data set can be found here: https://www.naturalearthdata.com/ downloads/110m-cultural-vectors/110m-populated-places/.
- 6. While GPC land sector guidance protocol (WRI 2022) is to use a 20-year amortization period for calculating emissions from land use changes, Pendrill et al. (2019a, 2019b, 2020) use a five-year amortization period for two main reasons. Firstly, the study is constrained by available data on tree cover loss to the period 2001–14. Thus, only 1-, 5-, and 10-year amortization periods were able to be compared. Differences between these periods were found to be negligible.
- 7. There are a few reasons why Pendrill et al. (2019a, 2019b, 2020) have omitted highly processed products from their analysis, the most salient of which is that Kastner et al. (2014a), on which Pendrill's report is based, focus on the trade of agricultural commodities irrespective of the differences of impact (i.e., deforestation) per country and region. Kastner et al. give the example of China's trade in cotton-based clothing. Because clothing is relatively equal in terms of imports and exports there is no need to track these types of products to estimate national consumption of cotton. Other factors such as the high degree of variation in the use of primary commodities in highly processed products (e.g., types of oils used in soaps) make tracking products using international trade data all the more difficult.
- See also Leblois et al. (2017), Ordoway et al. (2017), and the ongoing work of Global Forest Watch (https://www.globalforestwatch.org/) and geo-Footprint (https://geofootprint.com/) for other examples of ED estimation via remote sensing.
- Another method worth noting was developed by 3Keel Consultants in their 2017 study of the UK's national deforestation impact (WWF-UK and RSBP). They developed a "deforestation risk factor" on a scale of 1 to 12 for each commodity and source.

REFERENCES

Achard, F., R. Beuchle, P. Mayaux, H-J. Stibig, C. Bodart, A. Brink, S. Carboni, B. Desclée, F. Donnay, and H.D. Eva. 2014. "Determination of Tropical Deforestation Rates and Related Carbon Losses from 1990 to 2010." *Global Change Biology* 20 (8): 2540–54.

Baabou, W., N. Grunewald, C. Ouellet-Plamondon, M. Gressot, and A. Galli. 2017. "The Ecological Footprint of Mediterranean Cities: Awareness Creation and Policy Implications." *Environmental Science & Policy* 69: 94–104.

Baccini, A.G., S.J. Goetz, W.S. Walker, N.T. Laporte, M. Sun, D. Sulla-Menashe, J. Hackler, P.S.A. Beck, R. Dubayah, and M.A. Friedl. 2012. "Estimated Carbon Dioxide Emissions from Tropical Deforestation Improved by Carbon-Density Maps." *Nature Climate Change* 2 (3): 182–85.

Bontinck, P-A, T.F. Grant, M. Sevenster, S. Eady, and D. Crawford. 2020. "Improving Direct Land Use Change Calculations: An Australian Case Study." *International Journal of Life Cycle Assessment* 25: 998-1012.

Brugnoli, F., and I. Král'. 2012. "Life Cycle Assessment, Carbon Footprint in Leather Processing." Eighteenth Session of the Leather and Leather Products Industry Panel, Shanghai, China, September 1–5, 2012. Vienna: United Nations Industrial Development Organization (UNIDO).

Chapman, M., A. LaValle, G. Furey, and K.M.A. Chan. 2017. "Sustainability beyond City Limits: Can 'Greener' Beef Lighten a City's Ecological Footprint?" *Sustainability Science* 12 (4): 597–610.

Curtis, P.G., C. M. Slay, N.L. Harris, A. Tyukavina, and M.C. Hansen. 2018. "Classifying Drivers of Global Forest Loss." *Science* 361 (6407): 1108–11.

Davis, K.F., K. Yu, M.C. Rulli, L. Pichdara, and P. D'Odorico. 2015. "Accelerated Deforestation Driven by Large-Scale Land Acquisitions in Cambodia." *Nature Geoscience* 8 (10): 772–75.

Decision Innovation Solutions. 2021. 2020 Soybean Meal Demand Assessment. United Soybean Board. https://www.soymeal.org/2020soybean-meal-demand-assessment/.

European Commission. 2013. *The Impact of EU Consumption on Deforestation: Comprehensive Analysis of the Impact of EU Consumption on Deforestation: Final Report*, by D. Cuypers, A. Lust, T. Geerken, L. Gorissen, G. Peters, J. Karstensen, S. Prieler, et al. Luxembourg: European Commission. http://dx.publications.europa.eu/10.2779/82226.

EUROSTAT. 2021. "Household Budget Surveys." Luxembourg: European Commission. https://ec.europa.eu/eurostat/web/ household-budget-surveys/database. FAO (Food and Agricultural Organization of the United Nations). 2005. *Global Forest Resources Assessment 2005*. Rome: FAO.

FAO. 2018. "Global Forest Products: Facts and Figures 2018." Rome: FAO. http://www.fao.org/3/ca7415en/ca7415en.pdf.

FAO. 2020. "New Food Balances." Rome: FAO. http://www.fao.org/ faostat/en/#data/FBS/metadata. Accessed February 20, 2021.

FAO. n.d. FAOSTAT. Database. https://www.fao.org/faostat/en/#home.

FAO and World Bank. 2018. *Food Data Collection in Household Consumption and Expenditure Surveys: Guidelines for Low- and Middle-Income Countries.* Rome: FAO; Washington, DC: World Bank. https://openknowledge.worldbank.org/handle/10986/32503 License: CC BY-NC-SA 3.0 IGO.

Gac, A., C. Lapasin, P.T. Laspière, S. Guardia, P. Ponchant, P.Chevillon, and G. Nassy. 2014. "Co-products from Meat Processing: The Allocation Issue." *Proceedings of the 9th International Conference on Life Cycle Assessment in the Agri-Food Sector* (LCA Food 2014), San Francisco, CA, USA, October 8-10. 438–42.

GFW (Global Forest Watch). 2022. "Forest Pulse." Global Forest Review. Washington, DC: World Resource Institute. https://research.wri. org/gfr/latest-analysis-deforestation-trends?utm_campaign=treecoverloss2021&utm_medium=bitly&utm_source=GFWZoomEvent.

Goldman, E.D., M. Weisse, N. Harris, and M. Schneider. 2020. "Estimating the Role of Seven Commodities in Agriculture-Linked Deforestation: Oil Palm, Soy, Cattle, Wood Fiber, Cocoa, Coffee, and Rubber." Technical Note. Washington, DC: World Resources Institute.

Goldstein, B., M. Birkved, J. Fernández, and M. Hauschild. 2016. "Surveying the Environmental Footprint of Urban Food Consumption: The Urban Foodprint." *Journal of Industrial Ecology* 21 (1): 151–65.

Grand View Research. 2020. "Palm Oil Market Size, Share and Trends Analysis Report, by Origin (Organic, Conventional), by Product (Crude, RBD, Palm Kernel Oil, Fractionated), by End Use, by Region, and Segment Forecasts, 2020–2027." San Francisco: Grand View Research. https://www.grandviewresearch.com/industry-analysis/ palm-oil-market.

Hansen, M.C., P.V. Potapov, R. Moore, M. Hancher, S.A. Turubanova, A. Tyukavina, D. Thau, S.V. Stehman, S.J. Goetz, and T.R. Loveland. 2013. "High-Resolution Global Maps of 21st-Century Forest Cover Change." *Science* 342 (6160): 850–53.

Hansen, M.C., S.V. Stehman, and P.V. Potapov. 2010. "Quantification of Global Gross Forest Cover Loss." *Proceedings of the National Academy of Sciences* 107 (19): 8650–55.

Harris, N., E. D. Goldman, and S. Gibbes. 2019. "Spatial Database of Planted Trees Version 1.0." Technical Note. Washington, DC: World Resources Institute. https://www.wri.org/publication/planted-trees.

Heilmayr, R. 2014. "Conservation through Intensification? The Effects of Plantations on Natural Forests." *Ecological Economics* 105: 204–10.

Held, C., E. Meier-Landsberg, and V. Alonso. 2021. *Tropical Timber* 2050: An Analysis of the Future Supply of and Demand for Tropical Timber and Its Contributions to a Sustainable Economy. ITTO Technical Series No. 49. Yokohama, Japan: International Tropical Timber Organization (ITTO).

Hoang, N.T., and K. Kanemoto. 2021. "Mapping the Deforestation Footprint of Nations Reveals Growing Threat to Tropical Forests." *Nature Ecology & Evolution* 5 (6): 845–53. https://doi.org/10.1038/ s41559-021-01417-z.

ICO (International Coffee Organization). 2020. "World Coffee Consumption." London: ICO. http://www.ico.org/prices/new-consumption-table.pdf.

IFIF (International Feed Industry Federation). 2021. "IFIF Annual Report 2020/21." Luxembourg: IFIF. https://annualreport.ifif.org/

IPCC (Intergovernmental Panel on Climate Change). 2022. "Summary for Policymakers," edited by H.-O. Pörtner, D.C. Roberts, E.S. Poloczanska, K. Mintenbeck, M. Tignor, A. Alegría, M. Craig, S. Langsdorf, S. Löschke, V. Möller, A. Okem. In *Climate Change 2022: Impacts, Adaptation and Vulnerability.* Contribution of Working Group II to the Sixth Assessment Report of the Intergovernmental Panel on Climate Change, edited by H.-O. Pörtner, D.C. Roberts, M. Tignor, E.S. Poloczanska, K. Mintenbeck, A. Alegría, M. Craig, S. Langsdorf, S. Löschke, V. Möller, A. Okem, B. Rama. Cambridge and New York: Cambridge University Press. doi:10.1017/9781009325844.001.

Kastner, T., K-H. Erb, and H. Haberl. 2014a. "Rapid Growth in Agricultural Trade: Effects on Global Area Efficiency and the Role of Management." *Environmental Research Letters* 9 (3): 034015.

Kastner, T., K-H Erb, and S. Nonhebel. 2011a. "International Wood Trade and Forest Change: A Global Analysis." *Global Environmental Change* 21 (3): 947–56.

Kastner, T., M. Kastner, and S. Nonhebel. 2011b. "Tracing Distant Environmental Impacts of Agricultural Products from a Consumer Perspective." *Ecological Economics* 70 (6): 1032–40.

Kastner, T., A. Schaffartzik, N. Eisenmenger, K-H Erb, H. Haberl, and F. Krausmann. 2014b. "Cropland Area Embodied in International Trade: Contradictory Results from Different Approaches." *Ecological Economics* 104: 140–44.

Le Quéré, C., M.R. Raupach, J.G. Canadell, G. Marland, L. Bopp, P. Ciais, T.J. Conway, et al. 2009. "Trends in the Sources and Sinks of Carbon Dioxide." *Nature Geoscience* 2 (12): 831–36.

Leblois, A., O. Damette, and J. Wolfersberger. 2017. "What Has Driven Deforestation in Developing Countries since the 2000s? Evidence from New Remote-Sensing Data." *World Development* 92: 82–102.

Mokany, K., R.J. Raison, and A.S. Prokushkin. 2006. "Critical Analysis of Root:Shoot Ratios in Terrestrial Biomes." *Global Change Biology* 12 (1): 84–96.

Moore, J., M. Kissinger, and W.E. Rees. 2013. "An Urban Metabolism and Ecological Footprint Assessment of Metro Vancouver." *Journal of Environmental Management* 124: 51–61.

Ndikumana, L. 2016. "Trade Misinvoicing in Primary Commodities in Developing Countries: The Cases of Chile, Côte d'Ivoire, Nigeria, South Africa and Zambia." Geneva: United Nations Conference on Trade and Development.

OECD (Organisation for Economic Co-operation and Development). 2021. "OECD Stat: Metropolitan Areas." Database. https://stats.oecd. org/Index.aspx?DataSetCode=CITIES.

OECD and FAO. 2020. "OECD-FAO Agricultural Outlook 2020–2029." Paris and Rome: OECD Publishing and FAO. https://doi. org/10.1787/1112c23b-en.

Oil World. 2019. "Oil World Annual 2019." Hamburg, Germany: Oil World. https://www.oilworld.biz/t/publications/annual.

Opio, C., P. Gerber, A. Mottet, A. Falcucci, G. Tempio, M. MacLeod, T. Vellinga, B. Henderson, and H. Steinfeld. 2013. *Greenhouse Gas Emissions from Ruminant Supply Chains: A Global Life Cycle Assessment*. Rome: FAO.

Ordway, E.M., G.P. Asner, and E.F. Lambin. 2017. "Deforestation Risk Due to Commodity Crop Expansion in Sub-Saharan Africa." *Environmental Research Letters* 12 (4): 044015.

Pan, Y., R.A. Birdsey, J. Fang, R. Houghton, P.E. Kauppi, W.A. Kurz, O.L. Phillips, A. Shvidenko, S.L. Lewis, and J.G. Canadell. 2011. "A Large and Persistent Carbon Sink in the World's Forests." *Science* 333 (6045): 988–93.

Pendrill, F., U.M. Persson, J. Godar, and T. Kastner. 2019a. "Deforestation Displaced: Trade in Forest-Risk Commodities and the Prospects for a Global Forest Transition." *Environmental Research Letters* 14 (5): 055003. Pendrill, F., U.M. Persson, J. Godar, T. Kastner, D. Moran, S. Schmidt, and R. Wood. 2019b. "Agricultural and Forestry Trade Drives Large Share of Tropical Deforestation Emissions." *Global Environmental Change* 56: 1–10.

Pendrill, F., U.M. Persson, and T. Kastner. 2020. "Deforestation Risk Embodied in Production and Consumption of Agricultural and Forestry Commodities 2005–2017." Dataset. *Zenodo*. https://doi. org/10.5281/zenodo.4250532.

Potts, J., M. Lynch, A. Wilkings, G.A. Huppé, M. Cunningham, and V.A. Voora. 2014. *The State of Sustainability Initiatives Review 2014: Standards and the Green Economy*. Winnipeg, Canada: International Institute for Sustainable Development.

Proforest. 2011. "Mapping and Understanding the UK Palm Oil Supply Chain." London: Department for Environment, Food and Rural Affairs.

Ramankutty, N., H.K. Gibbs, F. Achard, R. Defries, J.A. Foley, and R.A. Houghton. 2006. "Challenges to Estimating Carbon Emissions from Tropical Deforestation." *Global Change Biology* 13 (1): 51–66.

Rees, W., and M. Wackernagel. 2008. "Urban Ecological Footprints: Why Cities Cannot Be Sustainable—and Why They Are a Key to Sustainability." In *Urban Ecology: An International Perspective on the Interactions between Humans and Nature*, edited by J.M. Marzluff, G. Bradley, E. Shulenberger, C. Ryan, W. Endlicher, U. Simon, M. Alberti, and C. Zumbrunner, 537–55. New York: Springer.

Roe, S., C. Streck, M. Obersteiner, S. Frank, B. Griscom, L. Drouet, O. Fricko, et al. 2019. "Contribution of the Land Sector to a 1.5 °C World." *Nature Climate Change* 9 (11): 817–28.

Saatchi, S.S., N.L. Harris, S. Brown, M. Lefsky, E.T.A. Mitchard, W. Salas, B.R. Zutta, W. Buermann, S.L. Lewis, and S. Hagen. 2011. "Benchmark Map of Forest Carbon Stocks in Tropical Regions across Three Continents." *Proceedings of the National Academy of Sciences* 108 (24): 9899–9904.

Searchinger, T.D., S. Wirsenius, T. Beringer, and P. Dumas. 2018. "Assessing the Efficiency of Changes in Land Use for Mitigating Climate Change." *Nature* 564: 249–53.

Seymour, F., and J. Busch. 2016. *Why Forests? Why Now? The Science, Economics, and Politics of Tropical Forests and Climate Change.* Washington, DC: Center for Global Development.

Statistics and Planning Department, Rubber Board. 2018. "The Year 2017[~]–18." *Rubber Statistical News Monthly* 76 (12). Kerala: India. http://rubberboard.org.in/rbfilereader?fileid=209.

Steinweg, T., B. Kuepper, and M. Piotrowski. 2019. "Palm Oil Biofuels Market May See Shake-up in 2020, Heightening Leakage Risks." Washington, DC: Chain Reaction Research. https://chainreactionresearch.com/report/palm-oil-biofuels-market-may-see-shake-up-in-2020-heightening-leakage-risks/.

UN (United Nations). 2015. UN Comtrade Database. New York: UN.

van Gelder, J., K. Kammeraat, and H. Kroes. 2008. "Soy Consumption for Feed and Fuel in the European Union." Castricum, The Netherlands: Profundo.

van Straaten, O., M.D. Corre, K. Wolf, M. Tchienkoua, E. Cuellar, R.B. Matthews, and E. Veldkamp. 2015. "Conversion of Lowland Tropical Forests to Tree Cash Crop Plantations Loses Up to One-Half of Stored Soil Organic Carbon." *Proceedings of the National Academy of Sciences* 112 (32): 9956–60.

Wackernagel, M., J. Kitzes, D. Moran, S. Goldfinger, and M. Thomas. 2006. "The Ecological Footprint of Cities and Regions: Comparing Resource Availability with Resource Demand." *Environment and Urbanization* 18 (1): 103–12.

Waite, R., D. Vennard, and G. Pozzi. 2019. "Tracking Progress toward the Cool Food Pledge: Setting Climate Targets, Tracking Metrics, Using the Cool Food Calculator, and Related Guidance for Pledge Signatories." Technical Note and Excel Calculator. Washington, DC: World Resources Institute.

Wolosin, M., and N. Harris. 2018. "Tropical Forests and Climate Change: The Latest Science." Washington, DC: World Resources Institute.

World Bank. 2021a. *Global Consumption Database*. Washington, DC: World Bank https://datatopics.worldbank.org/consumption/detail.

World Bank. 2021b. "Households and NPISHs Final Consumption Expenditure per Capita (Constant 2010 US\$)." https://data.worldbank. org/indicator/NE.CON.PRVT.PC.KD

WRI (World Resources Institute). *Resource Watch*. Washington, DC: WRI. https://resourcewatch.org/. Accessed February 20, 2021.

WRI, C40 Cities Climate Leadership Group, and ICLEI (Local Governments for Sustainability). 2021. *Global Protocol for Community-Scale Greenhouse Gas Emission Inventories*. Washington, DC: WRI. https://ghgprotocol.org/sites/default/files/standards/GPC_Full_ MASTER_RW_v7.pdf.

WRI, ICLEI-Local Governments for Sustainability, and C40 Cities Climate Leadership Group. 2022. *Global Protocol for Community-Scale Greenhouse Gas Inventories: Supplemental Guidance for Forests and Trees.* Washington, DC: WRI. WWF-UK and RSPB (Royal Society for Bird Protection). 2017. *Risky Business: Understanding the UK's Overseas Footprint for Deforestation-Risk Commodities*. Surrey and Sandy, UK: WWF-UK and RSPB._ https://www.wwf.org.uk/sites/default/files/2020-07/RiskyBusinessReport_2020.pdf.

Zarin, D.J., N.L. Harris, A. Baccini, D. Aksenov, M.C. Hansen, C. Azevedo-Ramos, T. Azevedo, et al. 2016. "Can Carbon Emissions from Tropical Deforestation Drop by 50% in 5 Years?" *Global Change Biology* 22 (4): 1336–47.

ABBREVIATIONS

COC	Carbon opportunity costs
COICOP	Classification of Individual Consumption by Purpose
ED	Embodied deforestation
FAO	Food and Agriculture Organization of the United Nations
GIS	Geographic information system
GHG	Greenhouse gas

ACKNOWLEDGMENTS

We are pleased to acknowledge our institutional strategic partners that provide core funding to WRI: the Netherlands Ministry of Foreign Affairs, Royal Danish Ministry of Foreign Affairs, and Swedish International Development Cooperation Agency.

The authors would like to thank Norway's International Climate and Forest Initiative (NICFI) and the United Kingdom's Department for Environment, Food & Rural Affairs (Defra) for their financial support to Cities4Forests, without which this report would not have been possible, and the Cities4Forests cofounding partners—Pilot Projects and REVOLVE—for their collaboration and support in creating the Cities4Forests initiative. The authors would like also to thank David Gibbs, Elizabeth Goldman, Nancy Harris, Cora Hallsworth, Eric Mackres, Ryan Mackie, Dan Moran, Nanda Noor, Florence Pendrill, Martin Persson, Richard Waite, Ted Wong, and Jessica Zionts for their thoughtful reviews of and feedback on the draft manuscript. Finally, we are grateful to Artemis Brod, Emily Matthews, Gregory Taff, and Romain Warnault for support during the editorial and production process.

ABOUT THE AUTHORS:

Mack Phillips is a Program Lead at Pilot Projects, with a focus on urban, exurban, and rural intersections. His prior research has covered topics such as the opportunity cost of optimal urbanization in suburban Montreal and the development of transit-oriented landscape amenities in exurban Sweden.

Scott Francisco is the Founder and Director of Pilot Projects. He is a designer and systems thinker with a focus on infrastructure that supports long-term cultural goals in cities, organizations, and ecosystems. He has a background in architecture, urbanism, and construction.

Sarah Jane Wilson is a Postdoctoral Fellow at the University of Victoria School of Environment and the Director of Nature-Based Projects at Pilot Projects. Her research and practice connect culture and ecology in landscapes urban and rural, including conservation and restoration strategies, education, and community engagement.

Paige Langer is an Associate with WRI's Global Forest Program, where her work focuses on tropical forests and their contribution to climate change mitigation and adaptation, as well as on the promotion of forests in international, domestic, and corporate climate strategies.

Larissa Jarvis is a Research Associate at the McGill University Sustainability Systems Initiative. Her research has covered topics such as land-use, agriculture, food security, and agricultural census data.

Noah Garcia is an educator, registered architect, multidisciplinary designer, and researcher focused on issues relating to the environment, technology, and education.

ABOUT WRI

World Resources Institute is a global research organization that turns big ideas into action at the nexus of environment, economic opportunity, and human well-being.

Our Challenge

Natural resources are at the foundation of economic opportunity and human well-being. But today, we are depleting Earth's resources at rates that are not sustainable, endangering economies and people's lives. People depend on clean water, fertile land, healthy forests, and a stable climate. Livable cities and clean energy are essential for a sustainable planet. We must address these urgent, global challenges this decade.

Our Vision

We envision an equitable and prosperous planet driven by the wise management of natural resources. We aspire to create a world where the actions of government, business, and communities combine to eliminate poverty and sustain the natural environment for all people.

Our Approach

COUNT IT

We start with data. We conduct independent research and draw on the latest technology to develop new insights and recommendations. Our rigorous analysis identifies risks, unveils opportunities, and informs smart strategies. We focus our efforts on influential and emerging economies where the future of sustainability will be determined.

CHANGE IT

We use our research to influence government policies, business strategies, and civil society action. We test projects with communities, companies, and government agencies to build a strong evidence base. Then, we work with partners to deliver change on the ground that alleviates poverty and strengthens society. We hold ourselves accountable to ensure our outcomes will be bold and enduring.

SCALE IT

We don't think small. Once tested, we work with partners to adopt and expand our efforts regionally and globally. We engage with decision-makers to carry out our ideas and elevate our impact. We measure success through government and business actions that improve people's lives and sustain a healthy environment.

©creative

Copyright 2022 World Resources Institute. This work is licensed under the Creative Commons Attribution 4.0 International License. To view a copy of the license, visit http://creativecommons.org/licenses/by/4.0/